Title: Common Genetic Variants on Chromosome 9p21 are Associated with Myocardial Infarction and Type 2 Diabetes in an Italian Population.

Authors:

Francesca Gori (francesca.gori@marionegri.it)
Claudia Specchia (specchia@med.unibs.it)
Silvia Pietri (silvia.pietri@marionegri.it)
Luisa Crociati (luisa.crociati@marionegri.it)
Simona Barlera (simona.barlera@marionegri.it)
Monica Franciosi (franciosi@negrisud.it)
Antonio Nicolucci (nicolucci@negrisud.it)
Stefano Signorini (signorini@uds.unimib.it)
Paolo Brambilla (paolo.brambilla@aovimercate.org)
Maria Grazia Franzosi (mariagrazia.franzosi@marionegri.it)

Version: 3 Date: 12 March 2010

Author's response to reviews: see over
To Scott Edmunds, PhD

Milan, 12th March 2010

Dear Dr Edmunds,

we submit our revised manuscript entitled “Common genetic variants on Chromosome 9p21 are associated with myocardial infarction and type 2 diabetes in an Italian population” for publication as research article in BMC Medical Genetics.

The manuscript has not already been published in any journal and all authors have read and agreed to its revised content.

As suggested by the Associate Editor we have done an English editing throughout and all changes are indicated with coloured underlined text in the present version of the manuscript we are submitting.

We hope the present version of the manuscript can be accepted for publication.

Sincerely yours

Dr. Francesca Gori,
Department of Cardiovascular Research
“Mario Negri” Institute for Pharmacological Research _Milan, Italy